

university of groningen

faculty of economics and business

2019004-EEF

Mental Health over the Life Course: Evidence for a U-Shape?

June 2019

Hermien Dijk Jochen Mierau

SOM is the research institute of the Faculty of Economics & Business at the University of Groningen. SOM has six programmes: - Economics, Econometrics and Finance - Global Economics & Management

- Innovation & Organization
- Marketing
- Operations Management & Operations Research
- Organizational Behaviour

Research Institute SOM Faculty of Economics & Business University of Groningen

Visiting address: Nettelbosje 2 9747 AE Groningen The Netherlands

Postal address: P.O. Box 800 9700 AV Groningen The Netherlands

T +31 50 363 7068/3815

www.rug.nl/feb/research



 faculty of economics and business

Mental Health over the Life Course: Evidence for a U-Shape?

Hermien Dijk University of Groningen, Faculty of Economics and Business, Department of Economics, Econometrics and Finance <u>h.h.dijk@rug.nl</u>

Jochen Mierau University of Groningen, Faculty of Economics and Business, Department of Economics, Econometrics and Finance j.o.mierau@rug.nl

Mental Health over the Life Course: Evidence for a U-Shape?*

May 30, 2019

Hermien Dijk[‡] University of Groningen

Jochen Mierau[‡] University of Groningen

Abstract

We aim to identify the age-profile of mental health while introducing minimal bias to reach identification. Using mental health data from the US Panel Study of Income Dynamics (PSID) we apply first difference estimation to derive an unbiased estimate of the second derivative of the age effect as well as an estimate up to a linear period trend of the first derivative. Next, we use a battery of estimators with varying restrictions to approximate the first derivative. We find conclusive evidence that the age profile of mental health in the US is not U-shaped and tentative evidence that the age-profile follows an inverse U-shape where individuals experience a mental health high during their life course. Further analyses, using German and Dutch data, confirm that these results do not only apply to the US, but also to Germany and to the Netherlands.

Keywords: Mental Health, Age-period-chort JEL Codes: I1, I18

^{*}In this paper use is made of data of the LISS (Longitudinal Internet Studies for the Social sciences) panel administered by CentERdata (Tilburg University, The Netherlands). Additionally we want to thank Rob Alessie, Michele Belloni and Swarnodeep Homroy for their invaluable comments. This research was financed by Accare. Apart from the funding, Accare was not involved in the analyses and the writing of this paper. Finally, the findings and views reported in this paper are those of the authors and should not be attributed to individuals or organisations mentioned here.

^bFaculty of Economics and Business, University of Groningen, P.O. Box 800, 9700 AV Groningen, The Netherlands. Phone: +31(0)50 363 3685, E-mail: h.h.dijk@rug.nl.

[‡]Faculty of Economics and Business, University of Groningen, P.O. Box 800, 9700 AV Groningen, The Netherlands. Phone: +31(0)50 363 3735, E-mail: j.o.mierau@rug.nl.

1 Introduction

Mental health problems are highly prevalent: approximately 20% of the working age population suffers from a mental disorder at any point in time and lifetime prevalence is estimated to be up to 50% (OECD, 2012). This high prevalence results both in an extremely large burden of disease (Murray et al., 2012; Whiteford et al., 2013), as well as significant societal costs: mental health problems are estimated to be the leading cause of years lived with disability worldwide (Whiteford et al., 2013) and the societal cost of mental disorders is estimated to be 3 to 4% of GDP in OECD countries (OECD, 2012, 2014). Therefore, it is important to know which groups are especially susceptible to mental health problems, so that interventions can be targeted at these groups.

One factor that appears to play a role in the burden of mental health problems is age. A number of studies have investigated the age-pattern of mental health (Bell, 2014; Blanchflower & Oswald, 2008, 2016; Le Bon & Le Bon, 2014; Lang, Llewellyn, Hubbard, Langa, & Melzer, 2011; Page, Milner, Morrell, & Taylor, 2013) and the majority of these studies have found that mental health follows a U-shaped pattern in age: young and old individuals generally experience better levels of mental health than individuals in, or close to, middle age (Blanchflower & Oswald, 2008, 2016; Le Bon & Le Bon, 2014; Lang et al., 2011). These results suggest that society would do well to invest more resources targeted at care and prevention of mental health problems among the middle aged.

All except one of these studies reporting U-shapes use cross-sectional evidence (i.e., Blanchflower & Oswald, 2016; Le Bon & Le Bon, 2014; Lang et al., 2011). However, investigating mental health trajectories over the life course is not without its caveats. By definition age, period and cohort are perfectly collinear (once an individual's age and the current year are known, it is possible to determine which year they were born). As a result, cross-sectional evidence of which age groups currently experience lower or higher mental health provides insufficient knowledge on the age-pattern of mental health, as it provides no evidence on whether the observed differences between age groups can be attributed to age effects or cohort effects. Hence, cross-sectional evidence cannot be generalized to future cohorts and age groups, and can provide no indication of whether interventions should either be targeted at specific cohorts or specific age groups. Because of the fundamental collinearity between age, period and cohort effects, statistical analysis on the subject requires assumptions with various degrees of arbitrariness that cannot be tested. A number of approaches has been suggested to tackle the problem, especially in the related literature on the age-effects of well being and life satisfaction, all with different assumptions and (dis)advantages. Unfortunately, these different approaches often lead to conflicting results. For example, studies assuming cohort effects are negligible consistently report U-shapes in mental health, life satisfaction or well-being (Blanchflower & Oswald, 2016, 2017; Graham & Pozuelo, 2017; Laaksonen, 2018; Le Bon & Le Bon, 2014; Lang et al., 2011), whereas studies on well-being assuming that period effects are negligible consistently report no U-shapes (FitzRoy, Nolan, Steinhardt, & Ulph, 2014; Frijters & Beatton, 2012; Kassenboehmer & Haisken-DeNew, 2012). Since the nature of the Age, Period and Cohort (APC) problem prevents formal testing of many of these assumptions the true age-profile remains unknown.

An alternative approach, proposed by De Ree and Alessie (2011) and Van Landeghem (2012) and used by Cheng, Powdthavee, and Oswald (2017), stands out because of its lack of need for arbitrary assumptions. By focusing on the first differences of life-satisfaction or well-being¹, these studies can identify age effects up to a linear trend. Hence, while the methods employed in these studies cannot prove the existence of a U-shape in mental health as individuals age (since the linear trend remains unknown), they can provide proof when the U-shape is nonexistent. The studies using this method generally find evidence supporting a U-shaped relationship between well-being (Van Landeghem, 2012; Cheng et al., 2017) or life satisfaction (De Ree & Alessie, 2011) and age.

Nevertheless, the inability of these studies to identify the linear age-trend is troubling, as this means that the true age-profile still remains unkown. To address this issue, Cheng et al. (2017) improve on the method applied by De Ree and Alessie (2011) and Van Landeghem (2012). However, their method has certain identification problems (for a more detailed explanation, see Appendix A). Therefore, the current study applies the method proposed by De Ree and Alessie (2011) and Van Landeghem (2012), but aims to provide more information on the linear age-effect by also providing the results of a battery of estimations using varying cohort restrictions.

¹Using this method they effectively estimate the second derivative with respect to age of a life-satisfaction/wellbeing equation.

We use data from three countries: the US Panel Study of Income Dynamics (PSID), as well as the German Socio-Economic Panel (SOEP) and the Dutch LISS panel. Our results indicate that the U-shape is not the dominant functional form in the relationship between mental health and age. In contrast, we find that the age-related profile of mental health likely follows an inverse U-shape, suggesting that the young and the elderly might be particularly at risk of developing mental health problems.

Aside from its relevance to mental healthcare and prevention policy, the methodological contribution of this paper to the literature is threefold. Firstly, to our knowledge, this paper is the first to apply the first difference approach proposed by De Ree and Alessie (2011) and Van Landeghem (2012) on mental health. As such, this study is the first to investigate the age-pattern of mental health without the necessity of arbitrary assumptions. Secondly, this paper introduces a relatively new approach regarding cohort restrictions. While it is not new to restrict a single cohort when using age, period and cohort effects as control variables, the current adaptation of the method - restriction across estimations - is new. Lastly, in contrast with a large fraction of the literature, this paper presents results from multiple estimation strategies for the linear age-effect. This is important, as different methodological choices appear to lead to different outcomes. Hence, just presenting one estimation strategy based on (a certain set of) arbitrary assumptions does not provide the complete picture.

This paper is structured as follows. Section 2 discusses the existing literature on the age-profile of mental health and well-being as well as the commonly applied methodologies. Section 3 provides a detailed explanation of the econometric methods used in this paper, after which Section 4 describes the data used for the main analysis. Results are provided in Section 5 and Section 6 contains the further analyses. Section 7 provides detailed discussion of the main result, after which section 8 provides a conclusion.

2 Background

Before delving into the literature on well-being, life satisfaction or mental health and age, it is important to understand the age-period-cohort (APC) problem and the proposed solutions to the problem. A much quoted (Bell & Jones, 2014, 2013, 2015) example of the distinction between age, period and cohort effects is the conversation between a senior worker (A) and a junior worker (B) from (Suzuki, 2012):

A: I can't seem to shake off this tired feeling. Guess I'm just getting old. [Age effect]

B: Do you think it's stress? Business is down this year, and you've let your fatigue build up. [Period effect]

A: Maybe. What about you?

B: Actually, I'm exhausted too! My body feels really heavy.

A: You're kidding. You're still young. I could work all day long when I was your age.

B: Oh, really?

A: Yeah, young people these days are quick to whine. We were not like that. [Cohort effect] (Suzuki, 2012).

The problem is that every combination of two items of this list perfectly predicts the third. In other words, age, period and cohort effects together exhibit perfect collinearity. As a result, they cannot be estimated in a standard regression equation. Many methodological solutions have been proposed to solve this problem, each with its own caveats.

We will not only report literature regarding the age-profile of mental health, but also the age-profile of life-satisfaction and well-being. These three concepts are closely related, as individuals with low life-satisfaction or well-being are more likely to report lower mental health and vice versa and as such, these strands of literature have been intertwined. Furthermore, the combined literature on life-satisfaction and well-being has been more extensive than the literature that focused exclusively on mental health.

Perhaps the simplest method to circumvent the APC problem is by simply assuming that either period, or cohort effects are negligible and can thus be ignored. Multiple studies have assumed that cohort effects are irrelevant (Blanchflower & Oswald, 2016, 2017; Graham & Pozuelo, 2017; Laaksonen, 2018; Le Bon & Le Bon, 2014; Lang et al., 2011) and all of these studies report U-shaped age profiles of mental health, life satisfaction or well-being. Others have instead assumed that period effects are irrelevant, which makes it possible to estimate age effects using fixed effects approaches (FitzRoy et al., 2014; Frijters & Beatton, 2012; Kassenboehmer & Haisken-DeNew, 2012; Piper, 2015). Most of these studies argue that there is no U-shaped relation between age and life satisfaction (FitzRoy et al., 2014; Frijters & Beatton, 2012; Kassenboehmer & Haisken-DeNew, 2012). The exception is provided by Piper (2015) whose results suggest that British individuals between 16 and 30 have a lower life satisfaction as they age.

Another method, which requires slightly weaker assumptions, is to assume that age, period and cohort effects are all relevant, but that either age groups, periods or cohorts close together have equal coefficients. E.g., using this assumption, one can run a regression using single-year age dummies, single-year period dummies, but five- or ten-year cohort dummies. Some studies have followed this approach (Blanchflower & Oswald, 2008; Lin, Hwang, & Deng, 2015; Page et al., 2013), but their results are mixed. Both Page et al. (2013) and Lin et al. (2015) provide no conclusive evidence regarding a U-shape in suicide rates and subjective wellbeing, respectively. Blanchflower and Oswald (2008) find evidence of a U-shape in well-being and rates of depression and anxiety.

Others have used Hierarchical APC (HAPC) models, which require the assumption that cohort and period effects are completely random (Yang, 2008; Bell, 2014; Beja, 2017). Yang (2008) and Bell (2014) use HAPC models and find no evidence of a U-shape in happiness and mental health, respectively. Beja (2017) applies a HAPC model to world wide life satisfaction data and does find a U-shaped relationship between age and life satisfaction for individuals aged 15-69.

The use of parameter restrictions has often been criticized (Bell & Jones, 2013, 2014; Luo & Hodges, 2016; De Ree & Alessie, 2011; O'Brien, 2017). By relying on arbitrary assumptions to reach identification, all of the studies above are likely to suffer from biases of unknown size. For example, by assuming cohort effects are negligible estimated age-effects include cohort-effects as well as age-effects.

To circumvent this issue, several authors/studies have attempted to identify age-effects using only minimal assumptions by focusing on the second derivative of the age profile (Cheng et al., 2017; De Ree & Alessie, 2011; Van Landeghem, 2012). The analyses in these studies can identify age patterns up to a linear trend; they can identify the second derivative of the age profile, but not the first.

This means that the true age-pattern remains unknown. For example, the second derivative of age might indicate the existence of a U-shape, but if the first derivative is sufficiently large (small), mental health, life statisfaction, or wellbeing is nevertheless continuously increasing (decreasing) over the life course. In such a case, a statistically significant second derivative only means that there is significant curvature in the downward (upward) sloping age-trend. Hence, while the methods employed in these studies cannot prove the existence of a U-shape in mental health as individuals age (since the linear trend remains unknown), they can prove that the U-shape is nonexistent.

The studies using this method generally find evidence supporting a U-shaped relationship between well-being (Van Landeghem, 2012; Cheng et al., 2017) or life satisfaction (De Ree & Alessie, 2011) and age. However, these results require caution as the exact functional form remains unknown.

Cheng et al. (2017) claim that they can identify the linear trend in age (i.e. the first derivative), by using a two-stage procedure where they first regress an equation for the first differences in life-satisfaction using only a set of year dummies as independent variables. They then use the residuals from this first regression as the dependent variable in the second stage regression that included a coefficient for age and a constant.

However, their analysis suffers from three problems. Firstly, it implicitly assumes that time trends are not linear. Consequently, in the presence of a linear time trend, their estimate of the first derivative cannot capture the linear time and age trend in the second stage as they have already been filtered out in the first stage. Secondly, since the first stage estimation ensures that the sum of the dependent variable in the second stage equals zero, the constant in the second stage is rather a product of the data structure than an estimator of the linear age-effect. Hence, the method applied by Cheng et al. (2017) cannot provide any additional information to the methods proposed by De Ree and Alessie (2011) and Van Landeghem (2012). Third, their coefficients are likely to be biased due to the fact that the variables for age and year are likely to be correlated, which they implicitly assume is not the case by running separate regressions for the two. For a more detailed explanation, see Appendix A.

What this overview of the literature teaches us is that there is no panacea when it comes to APC estimation. Any method for identifying APC effects either has a large probability of misspecification (parameter restrictions) or is underidentified (only inferring information about the second derivative). As a result, different methods can lead to different outcomes. This is clearly illustrated when studies assuming cohort effects are negligible are compared to studies assuming instead that period effects are negligible. The first category of studies consistently reported U-shapes, while the latter consistently reported no U-shapes. As a result, there is no consensus on the exact age-profile of mental health, life satisfaction and well-being.

With regard to mental health specifically, four of the six papers cited here do find a U-shaped age profiles of mental health variables (Blanchflower & Oswald, 2008, 2016; Le Bon & Le Bon, 2014; Lang et al., 2011). The fifth study (Page et al., 2013) provides no clear evidence in favour of or against the existence of a U-shape. However, this study only investigates the rather extreme case of suicide and is not necessarily focused on the existence of a U-shape.

The only study cited here that explicitly reports no U-shaped age profile in mental health uses the HAPC model (Bell, 2014). Bell (2014) argues that previous findings of a U-shape are the result from confounding of cohort effects and that, instead, mental health declines as individuals age.

In the sections that follow we investigate the age-profile of mental health using US panel data. To ensure that our results are not the artefacts of methodological choices, we will first estimate the second derivative using first differences, after which we will attempt to approximate the first derivative using a battery of parameter restrictions. Both methods will be explained in more detail below. Our results indicate that there is no U-shape in the relationship between mental health and age. In contrast, the relationship might even consist of an inverse U-shape.

3 Method

3.1 Second derivative

We base our analysis on Van Landeghem (2012) and De Ree and Alessie (2011) who both show that identification of the second derivative can be obtained by taking first differences from the dependent variable and regressing them on age as well as a set of year dummies². I.e., assume our Data Generating Process (DGP) is given by:

$$MH_{i,t} = \beta_0 + \beta_1 age_{i,t} + \beta_2 age_{i,t}^2 + \tau year_t + \gamma_t + \sum_{\phi=1905}^C \alpha_\phi cohort_i(\phi) + \epsilon_{i,t},$$
(1)

²Van Landeghem (2012) uses unaltered year dummies, De Ree and Alessie (2011) use Deaton Paxson year dummies (Deaton & Paxson, 1994). Both methods give equivalent estimates for the second derivative in our case.

where $MH_{i,t}$ denotes the mental health score of individual *i* in year *t*, $age_{i,t}$ denotes the age of individuals *i* in year *t*, τ is the parameter denoting a linear period effect and γ_t denotes the deviation from the linear period effect, $cohort_i(\phi)$ denotes a set of cohort dummies taking value one if ϕ equals the birth year of individual *i* and zero otherwise and $\epsilon_{i,t}$ denotes the error term. The parameters β_1 and β_2 are the parameters of interest, since they determine whether mental health is U-shaped over the life-course. By first differencing (1), and using the fact that $age_{i,t}^2 - age_{i,t-1}^2 = 2age_{i,t} - 1$, this can be rewritten as:

$$\Delta M H_{i,t} = (\beta_1 - \beta_2 + \tau) + (\gamma_t - \gamma_{t-1}) + 2\beta_2 age_{i,t} + \tilde{\epsilon}_{i,t}, \qquad (2)$$

where $\tilde{\epsilon}_{i,t} = \epsilon_{i,t} - \epsilon_{i,t-1}$. Hence, we can identify β_2 by using first differences of the mental health scores as dependent variables in a regression analysis with age multiplied by two and a set of year dummies as independent variables. We can only identify β_1 with this method if we are willing to make an assumption about the linear period trend τ .

3.2 First derivative

We will use parameter restrictions on cohorts to approximate the first derivative with respect to age. The least restrictive parameter restrictions model that allows for both linear age- and year-effects as well as nonlinear cohort-effects, but is still identified, is one where only two out of all cohorts are expected to have equal coefficients. This model is still biased, but the bias is minimized to the difference between the two cohort effects that are assumed to be equal. Additionally, if the average cohort effect approximates zero, on average our estimation of the linear age-effect should be close to the true coefficient.

Therefore, we estimate models of mental health using OLS with cluster robust standard errors with a continuous age variable, age squared, a set of year dummies and a set of (restricted) cohort dummies as independent variables. To reach identification the period dummies start from the second available year (i.e. 2001 is the reference year) and the cohort dummies consist of a full set of cohort dummies minus a reference cohort $(1902)^3$ and with the restriction that one cohort has a coefficient equal to the cohort from the previous birth year. ⁴

 $^{^{3}}$ grouping all cohorts with year of birth <1910 leads to similar results

 $^{^{4}}$ We do not use cohort restrictions on cohorts born before 1920 and after 1993, since these cohorts contain too few individuals (less than 100) to provide accurate results.

4 Data

We use data from the Panel Study of Income Dynamics (PSID) for the main analysis. The PSID started in 1968 and consists of a nationally representative sample of individuals living in families. From 2001 onwards, with the exception of 2005, the family interviews contained 6 questions belonging the abbreviated Kessler psychological distress scale (K-6) which were asked to the head and/or the partner of head of the household. The K-6 is a broad screener that can be used to assess non-sepcific psychological distress and the prevalence of serious mental illness in the general population (Kessler et al., 2002, 2010) and has been well validated (Prochaska, Sung, Max, Shi, & Ong, 2012; Furukawa et al., 2008; Furukawa, Kessler, Slade, & Andrews, 2003; Gill, Butterworth, Rodgers, & Mackinnon, 2007). We will use this variable as a measure of mental health. The family interviews were held biennially. Hence, in this study we use data from 9 family interviews, as 2017 data is not yet available and the 2005 interview did not contain the K-6 questions.

The K-6 consists of 6 questions asking how often individuals felt certain negative emotions during the last 30 days (sadness, nervous, restless, hopeless, that everything was an effort, worthless) which they can answer on a scale from 1 (all of the time) to 5 (none of the time) (Kessler et al., 2002). Generally these scores are inverted later so that a score of 0 means 'none of the time' and a score of 4 means 'all of the time' after which the scores from the different questions can be summed into a single score. For convenience, we linearly transformed this single score so that it ranges from 0 to 100, where 0 indicates that an individual answered 'all of the time' to all 6 questions.⁵

In our sample for analysis we included only the direct respondents of the family interview, as they were the ones answering the K-6 questions. Summary statistics of our final sample can be found in Table 1.⁶ Our first difference approach requires the presence of at least two consecutive observations of the K-6 for each individual in our set. As a result we lose 3,237 out of 14,378 individuals. However, there is almost no variation between both samples in terms of age, year of birth and K-6 score. Detailed summary statistics of this second sample

⁵Using the untransformed data does not alter the results.

⁶When split by gender the summary statistics show no substantial differences between men and women, see Tables 3 and 4 in Appendix B.1.

	Ν	Mean	Standard deviation	Minimum	Maximum
Age	$14,\!378$	38.7	15.99	16	99
Year of Birth	$14,\!378$	1965.84	18.10	1902	1997
Year	$14,\!378$	2005.08	4.92	2001	2015
Gender					
Female	$7,\!867$	54.72%			
Male	6,511	45.28%			
K-6 score (0-100)	$14,\!378$	83.93	17.27	0	100
Summary s	statistic a	t first obs	ervation (baseline) for	each individ	านอา

Table 1: Population summary statistics

at first observation (baseline) for each individual

can be found in Table 5 in Appendix B.1. Hence, we do not expect this to influence our results.

Figures 1, 2 and 3 provide a graphical analysis of the relation between mental health and age, period and cohort. In the graph, all years are pooled together and age-specific average K6 scores are stratified by 10-year cohort.⁷ It is interesting to note that our mental health measure does not show a strong U-pattern and that this is the case for both men and women. Mental health appears to increase at every age until individuals are around 70, after which at least for women - it declines slightly.

Another interesting observation is that visible cohort or period effects appear to be small: values for equal age groups of different cohorts are generally very similar. While this provides no conclusive evidence, it suggests that the patterns that we do see in the graphs might be due to age-effects.

5 Results

5.1Second derivative

To ensure that our estimates are robust to autocorrelation we have used cluster robust standard errors. Additionally, to reduce any effects of over- or undersampling in the PSID we performed our estimations with PSID family sample weights. Table 6 provides the results of the estimation of equation (2). It is interesting to note that our estimation results for

 $^{^{7}}$ Age groups within each cohort that consisted of less than 100 individuals were left out of the graph as the small samples resulted in volatile K6 scores for those age groups.



Figure 1: Average scores for: K6 (0-100 scale)

Figure 2: Scores women: K6 (0-100 scale)

Figure 3: Scores men: K6 (0-100 scale)



	Full sample	Women	Men
Δ K6 (0-100)			
$2age_{i,t}$	-0.009***	-0.009***	-0.009***
	(0.002)	(0.003)	(0.003)
Constant	1.203^{***}	1.545^{***}	0.734
	(0.313)	(0.408)	(0.489)
	100	TEC	TTO
Clustered SE	YES	YES	YES
Observations	$37,\!292$	$23,\!142$	$14,\!150$
Number of ID	$11,\!141$	6,737	4,404
Clustored	standard erro	re in naront	hosos

Table 2: Estimation results: second derivative

Clustered standard errors in parentheses *** p<0.01, ** p<0.05, * p<0.1

 β_2 do not vary across genders, both have a statistically significant estimate of -0.009. This indicates that there is no U-shape in mental health over the life course.

On the contrary, our estimation results for the second derivative with respect to age (β_2) indicate that there might be an inverse U-shape in mental health over age. However, whether this inverse U-shape truly exists depends on the linear term, which is not identified. If we are, however, willing to assume that there exists no linear time trend ($\tau = 0$), women's mental health would be continuously increasing until they reach old age (they reach their peak mental health at age 83), whereas men would experience a more profound inverse U-shape: given our current estimate, mental health would peak at age 39. Regardless of the size of the linear age effect, our finding of a negative second derivative disproves the existence of a U-shaped age profile in mental health.

5.2 First derivative

Figures 4-6 provide the estimated coefficients for the linear age effect for different cohort restrictions. For each restricted year of birth, the coefficient of that cohort is restricted to be equal to the coefficient of the previous cohort. I.e., if the restricted cohort is the one born in 1921, then the coefficient for the cohort effect of 1921 is assumed to be equal to that of 1920.

The parameter restrictions result in estimations of the linear age effect that vary between 0.412 and 0.673. Given these two extremes mental health would either peak somewhere between age 44 (linear age effect is 0.412) and age 72 (linear age effect is 0.673).



Figure 4: Estimates for linear age effect

Figure 5: Linear age effect: women

Figure 6: Linear age effect: men



When we stratify by gender, we find that the estimated linear age effects for women generally tend to be much higher than the linear age effects for men, suggesting that women reach their peak mental health later in life. For men, estimated linear age effects range from 0.237 to 0.538, suggesting mental health peaks sometime between age 25 and age 58. For women, the estimated coefficients vary between 0.343 and 0.774, suggesting that their mental health peaks sometime between age 37 and age 84. Apparently, the estimated ages at which individuals reach peak mental health using the varying cohort restrictions approach do not differ substantially from the estimated ages using the second derivative approach with the assumption of no linear period trend.

Consequently, if we are willing to assume that the bias caused by the parameter restrictions is either negligible, or at least close to zero on average, we can conclude that mental health is inversely U-shaped over the life course. This result applies to both men and women.

6 Further Analysis

6.1 Functional form

It could be that the DGP in this study is too restrictive. After all, in this study, as well as in the literature, there is no clear theoretical foundation for the assumption that mental health should either be linear or (inversely) U-shaped over age. Perhaps the age-profile of mental health is more complicated and, therefore, not approximated by the assumed DGP. It is relatively easy to test this hypothesis by assuming a less restrictive DGP. Following Van Landeghem (2012), we estimate equation (2) using a set of dummies for age instead of a single continuous age variable. The age-dependent second derivative is then given by differencing the coefficients of the age dummies. If the initial second order polynomial assumption is correct, the estimated age-dependent second derivative should only exhibit minimal variation around a straight line. If instead the pattern of the age-dependent second derivative is more complicated, this would suggest that our initial assumed DGP is incorrect.

A plot of the age-dependent second order derivatives is given in appendix B.2. As the graph clearly shows, the age-dependent derivative only shows minimal variation around a straight line. Additionally, a Ramsey RESET test (Ramsey, 1969) of equation (2) to assess the null hypothesis for no misspecification provides p-values of 0.035, 0.290 and 0.222 for the

entire sample, women and men, respectively. While the small p-value for the entire sample might suggest that the analysis suffers from some form of misspecification, the large p-values of the separate specifications for men and women suggest that this might largely be due to gender differences. Hence, the choice of a second order polynomial for age in equation (1) does not appear to cause misspecification. Consequently, the conclusions that the age-profile of mental health is not U-shaped and that it might even follow an inverse U-shape appear to be valid.

6.2 Attrition

There might be a possibility that individuals of certain ages with certain mental health levels are more likely to drop out of the sample. E.g., it might be that older individuals with more mental health problems have lower survival rates than older individuals with less mental health problems. If this is the case our estimate of the second derivative would be biased. However, when we add a dummy variable for attrition (indicating whether an individual will have left the panel in the next wave) to our second derivative estimation our results remain relatively unchanged (see Appendix B.3). Additionally, the coefficient for the attrition dummy is statistically insignificant (p > 0.10). Similarly, when we perform the second derivative estimation including only those individuals that were present in all waves, we find an estimate for the second derivative that is slightly higher, but still negative and statistically significantly different from zero (p < 0.05) and not statistically significantly different from our previous estimations (p > 0.10) (see Appendix B.3). Consequently, our results appear to be relatively unaffected by panel attrition.

6.3 Period and cohort trend

So far, we have provided two estimation strategies for the linear age-effect. The first estimation strategy (using first differences) required the assumption of a negligible period trend in order to identify the age-effect. The second strategy (using varying cohort restrictions) required the assumption of a negligible cohort trend. The degree of certainty about the first derivative hinges on the credibility of these assumptions. In other words, is there reason to believe either a linear period or cohort trend is more or less likely? In Appendix B.6 we provide two graphs: the first provides average K-6 (0-100) scores per year of observation and the second provides

average K-6 (0-100) scores per cohort. Note that both provide no conclusive evidence on the linear trend, as the averages are biased with age effects, and period or cohort effects. Nevertheless, average K-6 (0-100) scores show strikingly little variation over time, suggesting that perhaps the assumption of no linear period trend might not be too unrealistic.

6.4 Multiple Countries

Our finding that mental health is inversely U-shaped over the life course runs counter to the current literature, which more often than not reports a U-shape in mental health, wellbeing or life satisfaction. It might be though that this is simply due to the fact that we focus on the US, which might be the exception to the U-shaped rule (Blanchflower & Oswald, 2009). To test this assumption, we reperform our analysis of the second derivative on data from the Netherlands and Germany.

For our analysis of the Netherlans, we use data from the Dutch LISS panel ⁸, which consists of a representative sample of the Dutch population. The LISS panel contains data on individual mental health for the years 2007-2017, with the exception of 2014. Mental health is measured using the abbreviated 5-question version of the Mental Health Inventory (MHI-5) (Ware Jr & Sherbourne, 1992), which is a widely used and well validated instrument, specifically for mood and anxiety disorders (Veit & Ware, 1983; Rumpf, Meyer, Hapke, & John, 2001; McCabe, Thomas, Brazier, & Coleman, 1996). The MHI-5 can be summarized in a score ranging from 0 to 100, where higher scores indicate better mental health.

For Germany we use data from the German Socio-Economic Panel (SOEP). The SOEP is a long-run panel with data on German households from 1984 to 2016. A large number of papers have used SOEP to investigate a U-shape in life-satisfaction and well-being (e.g., Kassenboehmer & Haisken-DeNew, 2012; Frijters & Beatton, 2012; Van Landeghem, 2012; Baetschmann, 2014; FitzRoy et al., 2014; Cheng et al., 2017; Rohrer, Brümmer, Schupp, & Wagner, 2018; De Ree & Alessie, 2011), but, to our knowledge, currently no study has used SOEP to investigate the age-profile of mental health.

⁸The LISS panel is a representative sample of Dutch individuals who participate in monthly Internet surveys. The panel is based on a true probability sample of households drawn from the population register. Households that could not otherwise participate are provided with a computer and Internet connection. A longitudinal survey is fielded in the panel every year, covering a large variety of domains including work, education, income, housing, time use, political views, values and personality.

For mental health we use the Mental Component Summary scale (MCS), which is computed from answers to the SF-12v2 questionnaire using factor analysis (Andersen, Mühlbacher, Nübling, Schupp, & Wagner, 2007). The MCS is calibrated such that the population average is close to 50 and the standard deviation is close to 10, with higher scores implying better mental health. The MCS is available from 2002 onwards, hence we use data from all even years between 2002 and 2016, which resulted in data from 8 different years⁹

Estimated second derivatives for both datasets can be found in Table 7 of Appendix B.4. For Germany we find relatively similar results to the US, indicating that the non-existence of the U-shape in mental health in the US is not an isolated case. For the Netherlands we find a statistically insignificant second order derivative of 0.001 (SE: 0.002), indicating that either the second derivative is relatively close to 0 or our current assumed DGP might be misspecified for the Netherlands. We can test this last possibility in the same way as we did previously. Results for the age-dependent second derivative for the Netherlands can be found in Figure 8 of Appendix B.4. No clear age pattern emerges when looking at the agedependent second derivative for the Netherlands, suggesting that the second derivative might indeed be close to zero. Additionally, a Ramsey RESET test (Ramsey, 1969) could not reject the null hypothesis of no misspecification with a p-value of 0.174. Consequently, the U-shape in mental health is not only absent when using US data, but also when using German and Dutch data.

7 Discussion

Our results consistently indicate that there is no U-shape in mental health over age, and that this finding is not limited to just the US. This is not in line with the literature, which frequently reports a U-shape (Blanchflower & Oswald, 2008, 2016; Le Bon & Le Bon, 2014; Lang et al., 2011), which raises the question why our results differ.

One possible reason for the difference in results is that two of the four studies (Blanchflower & Oswald, 2016; Le Bon & Le Bon, 2014) use some form of mental healthcare use as proxies for mental health. The underlying assumption is that if the age profile of mental healthcare use consists of an inverse U-shape, the age profile of mental health must also be U-shaped. However, mental healthcare use might not be an adequate proxy for mental health and the

^{92002;2004;2006;2008;2010;2012;2014;2016}

reliability of healthcare use as a proxy might be dependent on age (Clement et al., 2015; Alonso et al., 2007; Wang et al., 2005). This could bias the results.

Additionally, three of the four studies do not control for cohort effects, assuming that these are zero (Blanchflower & Oswald, 2016; Le Bon & Le Bon, 2014; Lang et al., 2011). However, this might not be the case, leading to biased results. Even when cohort effects are taken into account (Blanchflower & Oswald, 2008), all four studies use superfluous parameter restrictions, suggesting that cohort effects are perhaps not adequately controlled for (Bell & Jones, 2013).

Lastly, many of these studies used control variables other than cohort and period variables to estimate the age effects of mental health (e.g. Blanchflower and Oswald (2008) and Lang et al. (2011), and to some extent Blanchflower and Oswald (2016)). This is not problematic per se, but it does beg the question what is being estimated exactly. It is easy to imagine that most of the age effect of mental health is not a direct result of individuals becoming older, but of other factors, physically and in society, that change as individuals age. In that sense adding control variables may seem like a good idea, but there is a vast number of factors that change as individuals age that potentially also affect mental health and it is difficult to asses in an analysis whether all relevant factors have been identified. As a result, when just a few of these factors are included in the analysis as control variables, it is unclear what the residual age effect found by the analysis consists of and, hence, it is difficult to interpret. Therefore, through the use of control variables, Blanchflower and Oswald (2008) and Lang et al. (2011) measure something fundamentally different (a residual age effect) from the crude age effect investigated in this study, which might lead to different results.

Indeed, when we perform an anlysis similar to Blanchflower and Oswald (2008), using 10-year cohort dummies and a variety of control variables we do find a U-shape in mental health, albeit a highly statistically insignificant one (see Appendix B.5). The combination of the two methodological differences (inadequate cohort controls and analysing a residual age-effect) force our previously negative, unbiased estimate for the second derivative to suddenly become positive. There is no clear interpretation that we can provide of this positive estimate, since it is unclear what it entails exactly as it combines both a cohort bias and an unknown residual age-effect. ¹⁰

 $^{^{10}}$ When we add these same control variables to the second derivative approach used in this paper, we still consistently find a negative second derivative, although it is no longer significant at a 5% or 1% level (results

Moreover, our results are in line with the results of Bell (2014) if we are willing to assume that there is no linear period effect. Bell (2014) implicitly makes this assumption by using a HAPC model, and just as in this study finds a negative coefficient for the second derivative.

In this study we have used a very narrow definition of mental health. The WHO (2014) has defined mental health as 'a state of well-being in which every individual realizes his or her own potential, can cope with the normal stresses of life, can work productively and fruitfully, and is able to make a contribution to her or his community.' While the questionnaires used in this study capture some of this definition of mental health, they certainly do not encompass all of the definition by the WHO (2014). Specifically, the K-6 focuses more on the presence or absence of psychological distress than the holistic view of mental health proposed by the WHO (2014). On the other hand, non-specific psychological distress is a common symptom in a broad range of mental disorders (Kessler et al., 2002) and all three instruments used in this study have evidence suggesting they can function as valid screening tools in the general population (Gill et al., 2007; Rumpf et al., 2001). Hence, while the current study might not necessarily reflect mental health in its entirety, it does capture important aspects of mental health.

Knowing the crude age pattern of mental health (or psychological distress) is highly important, as it indicates which age groups are at risk for mental health problems. A logical further step then is to see which variables drive this age pattern, so that perhaps specific policies and interventions can be targeted at improving the mental health of at-risk groups. To identify suspect variables, future studies could investigate which variables correlate with the age pattern of mental health, after which researchers can focus on identifying causality. Both of these steps are outside the scope of the current study.

From our background section we conclude that there is no panacea when it comes to APC estimation. Any method for identifying APC effects either has a large probability of misspecification or is underidentified. As a result, while we can derive an unbiased estimate for the second derivative, when estimating the first derivative different methods will lead to different biases and different outcomes. Consequently, any study reporting results from an APC analysis should be interpreted with caution. We have tried to circumvent this problem by widely varying the restrictions on cohorts used to reach identification and also reporting

available upon request). However, it should be noted that these control variables could introduce significant endogeneity bias into the estimate (Glenn, 2009).

the results of an estimation where the linear period-effect is assumed to be zero, assuming that if all these different restrictions give similar results our results are less likely to be the artefact of methodological choices. However, it is almost impossible to know in what direction results from different restrictions might be biased since that requires knowledge on all elements of the APC-problem which are by definition unidentifiable. As a result, we cannot know with certainty that the methods applied in this study do not suffer from biases as a result of a linear period or cohort trend. The only certainty we can provide is that various estimates with different restrictions are at least less likely to all suffer from a similar bias than a study employing just a single set of restrictions.

In other words, while we cannot be completely certain that the true linear age-effect lies within the bounds reported by this study, we have tried to reduce this uncertainty. Moreover, we can be certain that our estimate of the second derivative is unbiased. Hence, we can say with certainty that there is no U-shape in mental health over age and that this study provides tentative evidence that mental health follows an inverse U-shape over the life course.

8 Conclusion

This study investigated how mental health changes over the life-cycle by first employing an unbiased estimator for the second derivative of the age pattern, after which the linear age-effect was estimated by widely varying the restrictions on cohorts to reach identification. While a decent body of literature suggests that the age-profile of mental health might be U-shaped, we find evidence that this U-shape does not exist and might in fact be a methodological artefact. On the contrary, our results suggest that the relationship between mental health and age might actually be closer to an inverse U-shape, where individuals experience a mental health high at some point during their lives. This finding is highly societally relevant as it suggests that the young and the elderly might be particularly at risk for mental health problems. Future research should investigate what the determinants of the age pattern of mental health are.

References

- Alonso, J., Codony, M., Kovess, V., Angermeyer, M. C., Katz, S. J., Haro, J. M., ... others (2007). Population level of unmet need for mental healthcare in Europe. *The British Journal of Psychiatry*, 190(4), 299–306.
- Andersen, H. H., Mühlbacher, A., Nübling, M., Schupp, J., & Wagner, G. G. (2007). Computation of standard values for physical and mental health scale scores using the SOEP version of SF-12v2. Schmollers Jahrbuch, 127(1), 171–182.
- Baetschmann, G. (2014). Heterogeneity in the Relationship between Happiness and Age: Evidence from the German Socio-Economic Panel. German Economic Review, 15(3), 393–410.
- Beja, E. L. (2017). The U-shaped relationship between happiness and age: evidence using world values survey data. Quality & Quantity, 1–13.
- Bell, A. (2014). Life-course and cohort trajectories of mental health in the UK, 1991–2008–A multilevel age-period-cohort analysis. Social Science & Medicine, 120, 21–30.
- Bell, A., & Jones, K. (2013). The impossibility of separating age, period and cohort effects. Social Science & Medicine, 93, 163–165.
- Bell, A., & Jones, K. (2014). Another'futile quest'? A simulation study of Yang and Land's Hierarchical Age-Period-Cohort model. *Demographic Research*, 30, 333.
- Bell, A., & Jones, K. (2015). Age, period and cohort processes in longitudinal and life course analysis: A multilevel perspective. In A life course perspective on health trajectories and transitions (pp. 197–213). Springer.
- Blanchflower, D. G., & Oswald, A. (2017). Do Humans Suffer a Psychological Low in Midlife? Two Approaches (With and Without Controls) in Seven Data Sets. (Tech. Rep.). National Bureau of Economic Research.
- Blanchflower, D. G., & Oswald, A. J. (2008). Is well-being U-shaped over the life cycle? Social Science & Medicine, 66(8), 1733–1749.
- Blanchflower, D. G., & Oswald, A. J. (2009). The U-shape without controls: A response to Glenn. Social Science & Medicine, 69(4), 486–488.
- Blanchflower, D. G., & Oswald, A. J. (2016). Antidepressants and age: A new form of evidence for U-shaped well-being through life. *Journal of Economic Behavior & Organization*, 127, 46–58.

- Cheng, T. C., Powdthavee, N., & Oswald, A. J. (2017). Longitudinal Evidence for a Midlife Nadir in Human Well-being: Results from Four Data Sets. *The Economic Journal*, 127(599), 126–142.
- Clement, S., Schauman, O., Graham, T., Maggioni, F., Evans-Lacko, S., Bezborodovs, N., ... Thornicroft, G. (2015). What is the impact of mental health-related stigma on helpseeking? A systematic review of quantitative and qualitative studies. *Psychological Medicine*, 45(1), 11–27.
- Deaton, A. S., & Paxson, C. (1994). Saving, growth, and aging in Taiwan. In Studies in the economics of aging (pp. 331–362). University of Chicago Press.
- De Ree, J., & Alessie, R. (2011). Life satisfaction and age: Dealing with underidentification in age-period-cohort models. *Social Science & Medicine*, 73(1), 177–182.
- FitzRoy, F. R., Nolan, M. A., Steinhardt, M. F., & Ulph, D. (2014). Testing the tunnel effect: comparison, age and happiness in UK and German panels. *IZA Journal of European Labor Studies*, 3(1), 24.
- Frijters, P., & Beatton, T. (2012). The mystery of the U-shaped relationship between happiness and age. Journal of Economic Behavior & Organization, 82(2-3), 525–542.
- Furukawa, T. A., Kawakami, N., Saitoh, M., Ono, Y., Nakane, Y., Nakamura, Y., ... others (2008). The performance of the Japanese version of the K6 and K10 in the World Mental Health Survey Japan. International Journal of Methods in Psychiatric Research, 17(3), 152–158.
- Furukawa, T. A., Kessler, R. C., Slade, T., & Andrews, G. (2003). The performance of the K6 and K10 screening scales for psychological distress in the Australian National Survey of Mental Health and Well-Being. *Psychological Medicine*, 33(2), 357–362.
- Gill, S. C., Butterworth, P., Rodgers, B., & Mackinnon, A. (2007). Validity of the mental health component scale of the 12-item Short-Form Health Survey (MCS-12) as measure of common mental disorders in the general population. *Psychiatry Research*, 152(1), 63–71.
- Glenn, N. (2009). Is the apparent U-shape of well-being over the life course a result of inappropriate use of control variables? A commentary on Blanchflower and Oswald (66: 8, 2008, 1733–1749). Social Science & Medicine, 69(4), 481–485.

- Graham, C., & Pozuelo, J. R. (2017). Happiness, stress, and age: How the U curve varies across people and places. *Journal of Population Economics*, 30(1), 225–264.
- Kassenboehmer, S. C., & Haisken-DeNew, J. P. (2012). Heresy or enlightenment? The well-being age U-shape effect is flat. *Economics Letters*, 117(1), 235–238.
- Kessler, R. C., Andrews, G., Colpe, L. J., Hiripi, E., Mroczek, D. K., Normand, S.-L., ... Zaslavsky, A. M. (2002). Short screening scales to monitor population prevalences and trends in non-specific psychological distress. *Psychological Medicine*, 32(6), 959–976.
- Kessler, R. C., Green, J. G., Gruber, M. J., Sampson, N. A., Bromet, E., Cuitan, M., ... others (2010). Screening for serious mental illness in the general population with the K6 screening scale: results from the WHO World Mental Health (WMH) survey initiative. International Journal of Methods in Psychiatric Research, 19(S1), 4–22.
- Laaksonen, S. (2018). A Research Note: Happiness by Age is More Complex than U-Shaped. Journal of Happiness Studies, 19(2), 471–482.
- Lang, I., Llewellyn, D., Hubbard, R., Langa, K., & Melzer, D. (2011). Income and the midlife peak in common mental disorder prevalence. *Psychological Medicine*, 41(7), 1365–1372.
- Le Bon, O., & Le Bon, S.-D. (2014). Age distribution curve in psychiatric admissions inversely correlates with Life Satisfaction. *Psychiatry Research*, 219(1), 232–234.
- Lin, Y.-C., Hwang, R.-C., & Deng, W.-S. (2015). Heterogeneity in the relationship between subjective well-being and its determinants over the life cycle: A varying-coefficient ordered probit approach. *Economic Modelling*, 49, 372–386.
- Luo, L., & Hodges, J. S. (2016). Block Constraints in Age–Period–Cohort Models with Unequal-width Intervals. Sociological Methods & Research, 45(4), 700–726.
- McCabe, C., Thomas, K., Brazier, J., & Coleman, P. (1996). Measuring the mental health status of a population: A comparison of the GHQ-12 and the SF-36 (MHI-5). The British Journal of Psychiatry, 169(4), 517-521.
- Murray, C. J., Vos, T., Lozano, R., Naghavi, M., Flaxman, A. D., Michaud, C., ... others (2012). Disability-adjusted life years (dalys) for 291 diseases and injuries in 21 regions, 1990–2010: a systematic analysis for the global burden of disease study 2010. The lancet, 380(9859), 2197–2223.
- O'Brien, R. M. (2017). Mixed models, linear dependency, and identification in age-periodcohort models. *Statistics in Medicine*, 36(16), 2590–2600.

- OECD. (2012). Sick on the job?: myths and realities about mental health and work. OECD publishing Paris, France. doi: http://dx.doi.org/10.1787/9789264124523-en
- OECD. (2014). Making Mental Health Count: The Social and Economic Costs of Neglecting Mental Health Care. OECD publishing Paris, France. doi: 10.1787/9789264208445-en
- Page, A., Milner, A., Morrell, S., & Taylor, R. (2013). The role of under-employment and unemployment in recent birth cohort effects in Australian suicide. Social Science & Medicine, 93, 155–162.
- Piper, A. T. (2015). Sliding down the U-shape? A dynamic panel investigation of the age-well-being relationship, focusing on young adults. Social Science & Medicine, 143, 54–61.
- Prochaska, J. J., Sung, H.-Y., Max, W., Shi, Y., & Ong, M. (2012). Validity study of the K6 scale as a measure of moderate mental distress based on mental health treatment need and utilization. *International Journal of Methods in Psychiatric research*, 21(2), 88–97.
- Ramsey, J. B. (1969). Tests for specification errors in classical linear least-squares regression analysis. Journal of the Royal Statistical Society. Series B (Methodological), 350–371.
- Rohrer, J. M., Brümmer, M., Schupp, J., & Wagner, G. G. (2018). Worries across time and age in the German Socio-Economic Panel study. *Journal of Economic Behavior &* Organization.
- Rumpf, H.-J., Meyer, C., Hapke, U., & John, U. (2001). Screening for mental health: validity of the MHI-5 using DSM-IV Axis I psychiatric disorders as gold standard. *Psychiatry Research*, 105(3), 243–253.
- Suzuki, E. (2012). Time changes, so do people. Social Science & Medicine, 75(3), 452–456.
- Van Landeghem, B. (2012). A test for the convexity of human well-being over the life cycle: Longitudinal evidence from a 20-year panel. Journal of Economic Behavior & Organization, 81(2), 571–582.
- Veit, C. T., & Ware, J. E. (1983). The structure of psychological distress and well-being in general populations. *Journal of Consulting and Clinical Psychology*, 51(5), 730.
- Wang, P. S., Lane, M., Olfson, M., Pincus, H. A., Wells, K. B., & Kessler, R. C. (2005). Twelve-month use of mental health services in the United States: results from the

National Comorbidity Survey Replication. Archives of General Psychiatry, 62(6), 629–640.

- Ware Jr, J. E., & Sherbourne, C. D. (1992). The MOS 36-item short-form health survey (SF-36): I. Conceptual framework and item selection. *Medical Care*, 473–483.
- Whiteford, H. A., Degenhardt, L., Rehm, J., Baxter, A. J., Ferrari, A. J., Erskine, H. E., ... others (2013). Global burden of disease attributable to mental and substance use disorders: findings from the global burden of disease study 2010. *The Lancet*, 382(9904), 1575–1586.
- WHO. (2014). Mental health: a state of well-being. http://www.who.int/features/ factfiles/mental_health/en/#. ([Online; accessed 23-August-2018])
- Yang, Y. (2008). Social inequalities in happiness in the United States, 1972 to 2004: An age-period-cohort analysis. American Sociological Review, 73(2), 204–226.

Appendix

A Review of the method applied by Cheng, Powdthavee and Oswald (2017)

Cheng et al. (2017) investigate the age pattern of life satisfaction by examining a first difference equation with change-in-life-satisfaction as the dependent variable. To circumvent the problem of only being able to estimate the age-effect up to a linear trend, Cheng et al. (2017) state the following:

We circumvent it by using a simple two-step approach. In the first stage, we estimate a regression equation for change-in-life-satisfaction in which the only independent variables are a set of time dummies. Next, we calculate the residual from that equation; this residual is a measure of what might be called the de-trended change in life satisfaction for every person in the data set. We then use the residual as the dependent variable in the change-in-life-satisfaction regression equation. (Cheng et al., 2017)

Furthermore, in their appendix Cheng et al. (2017) provide the following assumed Data Generating Process (DGP) for the change-in-life-satisfaction equation with controls:

$$LS_{it} - LS_{it-1} = \Delta LS_{it} = \beta_0 + \beta_1 A_{it} + \theta P C_{it} + \mathbf{X}'_{it} \boldsymbol{\gamma} + \lambda T_t + \epsilon_{it},$$
(3)

where *i* and *t* index individual and time; LS_{it} is overall life satisfaction; A_{it} is individual's age; PC_{it} represents panel conditions or the length of time the individual is present in the panel at time *t*; \mathbf{X}'_{it} is a vector of standard socio-economic variables $[\ldots]$; T_t is time trend; and ϵ_{it} is the error term. (Cheng et al., 2017)

Since PC_{it} and X_{it} are part of the further analysis and are not an aspect of the main analysis, we will assume $\theta = 0$ and $\gamma = 0$. However, loosening these assumptions does not substantially alter the conclusions below.

Suppose there are four years of observations for $LS_{i,t}$ (i.e., t = 1, ..., 4), this would imply that there will be three years of observations for the first differences in equation (3) (i.e. t = 2, 3, 4). A regression with $\Delta LS_{i,t}$ as the dependent variable and a full set of year dummies as the independent variables would estimate the following:

$$\Delta LS_{i,2} = p_2 + u_{i,2},$$

$$\Delta LS_{i,3} = p_3 + u_{i,3},$$

$$\Delta LS_{i,4} = p_4 + u_{i,4},$$
(4)

where p_t is a year-specific constant and $u_{i,t}$ is the error term to be used as dependent variable in the second stage. Assuming momentarily that age and year are uncorrelated, it is easy to see that, given equation (3), $p_t = \beta_0 + \lambda t$. In other words, including a full set of year dummies filters the linear age trend (β_0 , the constant in the first-difference equation) out of the residuals ($u_{i,t}$) in the first stage and, therefore, it can never be identified in the second stage.

Even if we misinterpreted the two-way procedure and, in fact, one year-dummy was excluded from the set of equations (4) the linear age-effect would still be filtered out of all estimations of $u_{i,t}$ where there was a year dummy, and the estimations of $u_{i,t}$ without the year dummy would also include the linear period-effect next to the linear age-effect. Hence, even then, the linear age-effect cannot be identified in the second stage.

Additionally, by assuming β_0 denotes the linear age-effect of life satisfaction, Cheng et al. (2017) implicitly assume that there is no linear time-effect in the original life-satisfaction equation, since such a linear time-effect would result in a constant in the change-in-lifesatisfaction equation given by equation (3). Such a linear time-trend would indeed be captured by p_t , but as explained above, so would the linear age-effect.

One might wonder what the estimated constants assumed to represent the linear ageeffect in Cheng et al. (2017) consist of, if the method proposed by Cheng et al. (2017) cannot identify the linear age-effect. These constants are likely a product of the data structure and the estimated lows in midlife life-satisfaction are likely to equal the average age of all observations in the dataset. This is a result from the fact that age has not been centered, but the dependent variable (the estimated residuals from equation (4)) is. More specifically, since the first-stage estimation contains a constant (a full set of year dummies captures the constant), we know that:

$$\sum_{j=1}^{n} \hat{u}_n = 0, \tag{5}$$

where n is the total number of observations in the set and \hat{u}_n is the estimated residual from the first stage for every observation in the set. In the second stage Cheng et al. (2017) estimate the following regression model:

$$\hat{u}_n = \hat{\beta}_0 + \hat{\beta}_1 A_n + e_n. \tag{6}$$

Combining equation (5) and (6) yields

$$\sum_{j=1}^{n} \hat{u}_n = \sum_{j=1}^{n} (\hat{\beta}_0 + \hat{\beta}_1 A_n + e_n) = 0.$$
(7)

Since equation (6) also contains a constant $\sum_{j=1}^{n} e_n = 0$, hence we can derive that

$$\hat{\beta}_0 = -\hat{\beta}_1 \frac{\sum_{j=1}^n A_n}{n}.$$
(8)

The point at which equation (6) intersects with the X-axis is then given by

$$x = \frac{\sum_{j=1}^{n} A_n}{n},\tag{9}$$

i.e., the average age in the set. Hence, the fact that Cheng et al. (2017) consistently find a negative constant is to be expected given their estimation method. The fact that they find a minimum for life-satisfaction at middle age is also to be expected, since the average observation of age in the set is likely to be in, or close to, middle-age.

B Graphs and Figures

B.1 Summary statistics

	N	Mean	Standard deviation	Minimum	Maximum
Age	$7,\!867$	38.08	15.99	16	97
Year of Birth	$7,\!867$	1965.91	18.19	1903	1997
Year	$7,\!867$	2004.55	4.73	2001	2015
K-6 score $(0-100)$	$7,\!867$	83.18	17.39	0	100
~		0 1	. (1 1.) 0	1 . 1	

Table 3: Summary statistics: women

Summary statistic at first observation (baseline) for each individual

Table 4: Summary statistics: men

	Ν	Mean	Standard deviation	Minimum	Maximum
Age	6,511	39.38	15.97	16	99
Year of Birth	6,511	1965.76	18.00	1902	1997
Year	6,511	2005.72	5.07	2001	2015
K-6 score $(0-100)$	6,511	84.84	17.08	0	100
Summary statistic at first observation (baseline) for each individual					

Table 5	: Po	pulation	summary	statistics:	individuals	with at	least	two	consecutive	scores
		•	•/							

	Ν	Mean	Standard deviation	Minimum	Maximum	
Age	$11,\!141$	38.14	15.50	16	95	
Year of Birth	$11,\!141$	1965.18	17.52	1905	1995	
Year	$11,\!141$	2003.90	4.08	2001	2013	
Gender						
Female	6,737	60.47%				
Male	4,404	39.53%				
K-6 score $(0-100)$	$11,\!141$	84.53	16.69	0	100	
Current statistic at first absorbation (baseline) for each individual						

Summary statistic at first observation (baseline) for each individual

B.2 Age-dependent second derivative



Figure 7: Age-dependent second derivative

B.3 Attrition

Full sample	Individuals present in all waves
-	-
-0 009***	-0 007**
(0.002)	(0.003)
1.411***	1.231^{***}
(0.538)	(0.426)
-0.217	
(0.443)	
YES	YES
$37,\!292$	19,130
11,141	3,826
	Full sample -0.009*** (0.002) 1.411*** (0.538) -0.217 (0.443) YES 37,292 11,141

Table 6: Estimation results: second derivative

Clustered standard errors in parentheses *** p<0.01, ** p<0.05, * p<0.1

B.4 Netherlands and Germany

	Netherlands	Germany					
$2age_{i,t}$	0.001	-0.006***					
	(0.002)	(0.001)					
Constant	0.373	0.793^{***}					
	(0.267)	(0.137)					
Clustered SE	YES	YES					
Sample weights		YES					
Observations	$33,\!511$	111,706					
Number of ID	$9,\!250$	$35,\!286$					
Cluster robust standard errors in parentheses							
*** p<0.01, ** p<0.05, * p<0.1							

Table 7: Estimation results: second derivative

Figure 8: Age-dependent second derivative: Netherla	anc	ls
---	-----	----



Second derivative estimates for individuals aged >90 are not shown due to extreme outliers as a result of limited sample sizes for these age groups.

B.5 Residual age-effect with 10-year cohorts

K6 (0-100)	
$age_{i,t}$	0.007
	(0.096)
$age_{i,t}^2$	0.001
-,-	(0.001)
Cohort dummies	10-year
Year dummies	1-year
Controls	YES
Clustered SE	YES
Sample weights	YES
Observations	40,808
Number of ID	$11,\!695$
Control variables consisted of state dum	mies, the natural
log of income, marital status dummies, e	ducation level
dummies, employment status dummies,	race dummies and

a dummy variable denoting whether the household included

Table 8: Estimation results: residual age-effect with 10-year cohorts

children under 18. Cluster robust standard errors in parentheses *** p<0.01, ** p<0.05, * p<0.1

B.6 Cohort and period trends



Figure 9: Average K6 scores per year (0-100)



Figure 10: Average K6 scores per cohort (0-100)

List of research reports

14001-OPERA: Germs, R. and N.D. van Foreest, Optimal control of production-inventory systems with constant and compound poisson demand

14002-EEF: Bao, T. and J. Duffy, Adaptive vs. eductive learning: Theory and evidence

14003-OPERA: Syntetos, A.A. and R.H. Teunter, On the calculation of safety stocks

14004-EEF: Bouwmeester, M.C., J. Oosterhaven and J.M. Rueda-Cantuche, Measuring the EU value added embodied in EU foreign exports by consolidating 27 national supply and use tables for 2000-2007

14005-OPERA: Prak, D.R.J., R.H. Teunter and J. Riezebos, Periodic review and continuous ordering

14006-EEF: Reijnders, L.S.M., The college gender gap reversal: Insights from a life-cycle perspective

14007-EEF: Reijnders, L.S.M., Child care subsidies with endogenous education and fertility

14008-EEF: Otter, P.W., J.P.A.M. Jacobs and A.H.J. den Reijer, A criterion for the number of factors in a data-rich environment

14009-EEF: Mierau, J.O. and E. Suari Andreu, Fiscal rules and government size in the European Union

14010-EEF: Dijkstra, P.T., M.A. Haan and M. Mulder, Industry structure and collusion with uniform yardstick competition: theory and experiments

14011-EEF: Huizingh, E. and M. Mulder, Effectiveness of regulatory interventions on firm behavior: a randomized field experiment with e-commerce firms

14012-GEM: Bressand, A., Proving the old spell wrong: New African hydrocarbon producers and the 'resource curse'

14013-EEF: Dijkstra P.T., Price leadership and unequal market sharing: Collusion in experimental markets

14014-EEF: Angelini, V., M. Bertoni, and L. Corazzini, Unpacking the determinants of life satisfaction: A survey experiment

14015-EEF: Heijdra, B.J., J.O. Mierau, and T. Trimborn, Stimulating annuity markets

14016-GEM: Bezemer, D., M. Grydaki, and L. Zhang, Is financial development bad for growth?

14017-EEF: De Cao, E. and C. Lutz, Sensitive survey questions: measuring attitudes regarding female circumcision through a list experiment

14018-EEF: De Cao, E., The height production function from birth to maturity



faculty of e and busine

14019-EEF: Allers, M.A. and J.B. Geertsema, The effects of local government amalgamation on public spending and service levels. Evidence from 15 years of municipal boundary reform

14020-EEF: Kuper, G.H. and J.H. Veurink, Central bank independence and political pressure in the Greenspan era

14021-GEM: Samarina, A. and D. Bezemer, Do Capital Flows Change Domestic Credit Allocation?

14022-EEF: Soetevent, A.R. and L. Zhou, Loss Modification Incentives for Insurers Under ExpectedUtility and Loss Aversion

14023-EEF: Allers, M.A. and W. Vermeulen, Fiscal Equalization, Capitalization and the Flypaper Effect.

14024-GEM: Hoorn, A.A.J. van, Trust, Workplace Organization, and Comparative Economic Development.

14025-GEM: Bezemer, D., and L. Zhang, From Boom to Bust in de Credit Cycle: The Role of Mortgage Credit.

14026-GEM: Zhang, L., and D. Bezemer, How the Credit Cycle Affects Growth: The Role of Bank Balance Sheets.

14027-EEF: Bružikas, T., and A.R. Soetevent, Detailed Data and Changes in Market Structure: The Move to Unmanned Gasoline Service Stations.

14028-EEF: Bouwmeester, M.C., and B. Scholtens, Cross-border Spillovers from European Gas Infrastructure Investments.

14029-EEF: Lestano, and G.H. Kuper, Correlation Dynamics in East Asian Financial Markets.

14030-GEM: Bezemer, D.J., and M. Grydaki, Nonfinancial Sectors Debt and the U.S. Great Moderation.

14031-EEF: Hermes, N., and R. Lensink, Financial Liberalization and Capital Flight: Evidence from the African Continent.

14032-OPERA: Blok, C. de, A. Seepma, I. Roukema, D.P. van Donk, B. Keulen, and R. Otte, Digitalisering in Strafrechtketens: Ervaringen in Denemarken, Engeland, Oostenrijk en Estland vanuit een Supply Chain Perspectief.

14033-OPERA: Olde Keizer, M.C.A., and R.H. Teunter, Opportunistic condition-based maintenance and aperiodic inspections for a two-unit series system.

14034-EEF: Kuper, G.H., G. Sierksma, and F.C.R. Spieksma, Using Tennis Rankings to Predict Performance in Upcoming Tournaments

15001-EEF: Bao, T., X. Tian, X. Yu, Dictator Game with Indivisibility of Money



 faculty of economi and business

15002-GEM: Chen, Q., E. Dietzenbacher, and B. Los, The Effects of Ageing and Urbanization on China's Future Population and Labor Force

15003-EEF: Allers, M., B. van Ommeren, and B. Geertsema, Does intermunicipal cooperation create inefficiency? A comparison of interest rates paid by intermunicipal organizations, amalgamated municipalities and not recently amalgamated municipalities

15004-EEF: Dijkstra, P.T., M.A. Haan, and M. Mulder, Design of Yardstick Competition and Consumer Prices: Experimental Evidence

15005-EEF: Dijkstra, P.T., Price Leadership and Unequal Market Sharing: Collusion in Experimental Markets

15006-EEF: Anufriev, M., T. Bao, A. Sutin, and J. Tuinstra, Fee Structure, Return Chasing and Mutual Fund Choice: An Experiment

15007-EEF: Lamers, M., Depositor Discipline and Bank Failures in Local Markets During the Financial Crisis

15008-EEF: Oosterhaven, J., On de Doubtful Usability of the Inoperability IO Model

15009-GEM: Zhang, L. and D. Bezemer, A Global House of Debt Effect? Mortgages and Post-Crisis Recessions in Fifty Economies

15010-I&O: Hooghiemstra, R., N. Hermes, L. Oxelheim, and T. Randøy, The Impact of Board Internationalization on Earnings Management

15011-EEF: Haan, M.A., and W.H. Siekman, Winning Back the Unfaithful while Exploiting the Loyal: Retention Offers and Heterogeneous Switching Costs

15012-EEF: Haan, M.A., J.L. Moraga-González, and V. Petrikaite, Price and Match-Value Advertising with Directed Consumer Search

15013-EEF: Wiese, R., and S. Eriksen, Do Healthcare Financing Privatisations Curb Total Healthcare Expenditures? Evidence from OECD Countries

15014-EEF: Siekman, W.H., Directed Consumer Search

15015-GEM: Hoorn, A.A.J. van, Organizational Culture in the Financial Sector: Evidence from a Cross-Industry Analysis of Employee Personal Values and Career Success

15016-EEF: Te Bao, and C. Hommes, When Speculators Meet Constructors: Positive and Negative Feedback in Experimental Housing Markets

15017-EEF: Te Bao, and Xiaohua Yu, Memory and Discounting: Theory and Evidence

15018-EEF: Suari-Andreu, E., The Effect of House Price Changes on Household Saving Behaviour: A Theoretical and Empirical Study of the Dutch Case

15019-EEF: Bijlsma, M., J. Boone, and G. Zwart, Community Rating in Health Insurance: Trade-off between Coverage and Selection



15020-EEF: Mulder, M., and B. Scholtens, A Plant-level Analysis of the Spill-over Effects of the German *Energiewende*

15021-GEM: Samarina, A., L. Zhang, and D. Bezemer, Mortgages and Credit Cycle Divergence in Eurozone Economies

16001-GEM: Hoorn, A. van, How Are Migrant Employees Manages? An Integrated Analysis

16002-EEF: Soetevent, A.R., Te Bao, A.L. Schippers, A Commercial Gift for Charity

16003-GEM: Bouwmeerster, M.C., and J. Oosterhaven, Economic Impacts of Natural Gas Flow Disruptions

16004-MARK: Holtrop, N., J.E. Wieringa, M.J. Gijsenberg, and P. Stern, Competitive Reactions to Personal Selling: The Difference between Strategic and Tactical Actions

16005-EEF: Plantinga, A. and B. Scholtens, The Financial Impact of Divestment from Fossil Fuels

16006-GEM: Hoorn, A. van, Trust and Signals in Workplace Organization: Evidence from Job Autonomy Differentials between Immigrant Groups

16007-EEF: Willems, B. and G. Zwart, Regulatory Holidays and Optimal Network Expansion

16008-GEF: Hoorn, A. van, Reliability and Validity of the Happiness Approach to Measuring Preferences

16009-EEF: Hinloopen, J., and A.R. Soetevent, (Non-)Insurance Markets, Loss Size Manipulation and Competition: Experimental Evidence

16010-EEF: Bekker, P.A., A Generalized Dynamic Arbitrage Free Yield Model

16011-EEF: Mierau, J.A., and M. Mink, A Descriptive Model of Banking and Aggregate Demand

16012-EEF: Mulder, M. and B. Willems, Competition in Retail Electricity Markets: An Assessment of Ten Year Dutch Experience

16013-GEM: Rozite, K., D.J. Bezemer, and J.P.A.M. Jacobs, Towards a Financial Cycle for the US, 1873-2014

16014-EEF: Neuteleers, S., M. Mulder, and F. Hindriks, Assessing Fairness of Dynamic Grid Tariffs

16015-EEF: Soetevent, A.R., and T. Bružikas, Risk and Loss Aversion, Price Uncertainty and the Implications for Consumer Search

16016-HRM&OB: Meer, P.H. van der, and R. Wielers, Happiness, Unemployment and Self-esteem

16017-EEF: Mulder, M., and M. Pangan, Influence of Environmental Policy and Market Forces on Coal-fired Power Plants: Evidence on the Dutch Market over 2006-2014

16018-EEF: Zeng,Y., and M. Mulder, Exploring Interaction Effects of Climate Policies: A Model Analysis of the Power Market

16019-EEF: Ma, Yiqun, Demand Response Potential of Electricity End-users Facing Real Time Pricing

16020-GEM: Bezemer, D., and A. Samarina, Debt Shift, Financial Development and Income Inequality in Europe

16021-EEF: Elkhuizen, L, N. Hermes, and J. Jacobs, Financial Development, Financial Liberalization and Social Capital

16022-GEM: Gerritse, M., Does Trade Cause Institutional Change? Evidence from Countries South of the Suez Canal

16023-EEF: Rook, M., and M. Mulder, Implicit Premiums in Renewable-Energy Support Schemes

17001-EEF: Trinks, A., B. Scholtens, M. Mulder, and L. Dam, Divesting Fossil Fuels: The Implications for Investment Portfolios

17002-EEF: Angelini, V., and J.O. Mierau, Late-life Health Effects of Teenage Motherhood

17003-EEF: Jong-A-Pin, R., M. Laméris, and H. Garretsen, Political Preferences of (Un)happy Voters: Evidence Based on New Ideological Measures

17004-EEF: Jiang, X., N. Hermes, and A. Meesters, Financial Liberalization, the Institutional Environment and Bank Efficiency

17005-EEF: Kwaak, C. van der, Financial Fragility and Unconventional Central Bank Lending Operations

17006-EEF: Postelnicu, L. and N. Hermes, The Economic Value of Social Capital

17007-EEF: Ommeren, B.J.F. van, M.A. Allers, and M.H. Vellekoop, Choosing the Optimal Moment to Arrange a Loan

17008-EEF: Bekker, P.A., and K.E. Bouwman, A Unified Approach to Dynamic Mean-Variance Analysis in Discrete and Continuous Time

17009-EEF: Bekker, P.A., Interpretable Parsimonious Arbitrage-free Modeling of the Yield Curve

17010-GEM: Schasfoort, J., A. Godin, D. Bezemer, A. Caiani, and S. Kinsella, Monetary Policy Transmission in a Macroeconomic Agent-Based Model

17011-I&O: Bogt, H. ter, Accountability, Transparency and Control of Outsourced Public Sector Activities

17012-GEM: Bezemer, D., A. Samarina, and L. Zhang, The Shift in Bank Credit Allocation: New Data and New Findings

17013-EEF: Boer, W.I.J. de, R.H. Koning, and J.O. Mierau, Ex-ante and Ex-post Willingness-to-pay for Hosting a Major Cycling Event

17014-OPERA: Laan, N. van der, W. Romeijnders, and M.H. van der Vlerk, Higher-order Total Variation Bounds for Expectations of Periodic Functions and Simple Integer Recourse Approximations

17015-GEM: Oosterhaven, J., Key Sector Analysis: A Note on the Other Side of the Coin

17016-EEF: Romensen, G.J., A.R. Soetevent: Tailored Feedback and Worker Green Behavior: Field Evidence from Bus Drivers

17017-EEF: Trinks, A., G. Ibikunle, M. Mulder, and B. Scholtens, Greenhouse Gas Emissions Intensity and the Cost of Capital

17018-GEM: Qian, X. and A. Steiner, The Reinforcement Effect of International Reserves for Financial Stability

17019-GEM/EEF: Klasing, M.J. and P. Milionis, The International Epidemiological Transition and the Education Gender Gap

2018001-EEF: Keller, J.T., G.H. Kuper, and M. Mulder, Mergers of Gas Markets Areas and Competition amongst Transmission System Operators: Evidence on Booking Behaviour in the German Markets

2018002-EEF: Soetevent, A.R. and S. Adikyan, The Impact of Short-Term Goals on Long-Term Objectives: Evidence from Running Data

2018003-MARK: Gijsenberg, M.J. and P.C. Verhoef, Moving Forward: The Role of Marketing in Fostering Public Transport Usage

2018004-MARK: Gijsenberg, M.J. and V.R. Nijs, Advertising Timing: In-Phase or Out-of-Phase with Competitors?

2018005-EEF: Hulshof, D., C. Jepma, and M. Mulder, Performance of Markets for European Renewable Energy Certificates

2018006-EEF: Fosgaard, T.R., and A.R. Soetevent, Promises Undone: How Committed Pledges Impact Donations to Charity

2018007-EEF: Durán, N. and J.P. Elhorst, A Spatio-temporal-similarity and Common Factor Approach of Individual Housing Prices: The Impact of Many Small Earthquakes in the North of Netherlands

2018008-EEF: Hermes, N., and M. Hudon, Determinants of the Performance of Microfinance Institutions: A Systematic Review

2018009-EEF: Katz, M., and C. van der Kwaak, The Macroeconomic Effectiveness of Bank Bail-ins

university of groningen

2018010-OPERA: Prak, D., R.H. Teunter, M.Z. Babai, A.A. Syntetos, and J.E. Boylan, Forecasting and Inventory Control with Compound Poisson Demand Using Periodic Demand Data

2018011-EEF: Brock, B. de, Converting a Non-trivial Use Case into an SSD: An Exercise

2018012-EEF: Harvey, L.A., J.O. Mierau, and J. Rockey, Inequality in an Equal Society

2018013-OPERA: Romeijnders, W., and N. van der Laan, Inexact cutting planes for twostage mixed-integer stochastic programs

2018014-EEF: Green, C.P., and S. Homroy, Bringing Connections Onboard: The Value of Political Influence

2018015-OPERA: Laan, N. van der, and W. Romeijnders, Generalized aplhaapproximations for two-stage mixed-integer recourse models

2018016-GEM: Rozite, K., Financial and Real Integration between Mexico and the United States

2019001-EEF: Lugalla, I.M., J. Jacobs, and W. Westerman, Drivers of Women Entrepreneurs in Tourism in Tanzania: Capital, Goal Setting and Business Growth

2019002-EEF: Brock, E.O. de, On Incremental and Agile Development of (Information) Systems

2019003-OPERA: Laan, N. van der, R.H. Teunter, W. Romeijnders, and O.A. Kilic, The Data-driven Newsvendor Problem: Achieving On-target Service Levels.

2019004-EEF: Dijk, H., and J. Mierau. Mental Health over the Life Course: Evidence for a U-Shape?

www.rug.nl/feb